Author's response to reviews

Title: Abdominal Surgical Site Infections: Incidence and Risk Factors at an Iranian teaching hospital

Authors:

S. Mansour Razavi (razavy@sina.tums.ac.ir)
Mohammad Ibrahimpoor (ibrahimpoor@sina.tums.ac.ir)
S. Ali Jaffarian (jaffarian@sina.tums.ac.ir)
Ahmad Sabouri Kashani (kashani45@yahoo.com)

Version: 2 Date: 9 August 2004

Author’s response to reviews:

Abdominal Surgical Site Infections: Incidence and Risk Factors at an Iranian Teaching Hospital
Authors: S. Mansour Razavi, associate professor of community medicine, school of medicine, Tehran University of Medical Sciences (TUMS), Mohammad Ibrahimpoor, M.D., S. Ali Jaffarian, assistant prof., Surgery department, and Ahmad Sabouri Kashani, assistant prof., Educational Development Center, TUMS.

Person to be communicated: Dr. S. Mansour Razavi,
Dean of clinical education
School of Medicine, TUMS.
e-mail: razavy@sina.tums.ac.ir